Author's response to reviews

Title: Postoperative phlegmasia caerulea dolens: Potential iatrogenic factors.

Authors:

Ronan A Cahill (rcahill@rcsi.ie)
Henry P Redmond (cahillronan@hotmail.com)

Version: 2 Date: 10 September 2007

Author's response to reviews:

Written consent was received from the appropriate next of kin: